Author’s response to reviews

Title: PD-L1 assessment in pediatric rhabdomyosarcoma: a pilot study.

Authors:

Giulia Bertolini (giulia.bertolini@istitutotumori.mi.it)
Luca Bergamaschi (luca.bergamaschi@istitutotumori.mi.it)
Andrea Ferrari (andrea.ferrari@istitutotumori.mi.it)
Salvatore Renne (salvatore.renne@humanitas.it)
Paola Collini (paola.collini@istitutotumori.mi.it)
Cecilia Gardelli (cecg@libero.it)
Marta Barisella (marta.barisella@istitutotumori.mi.it)
Giovanni Centonze (giovanni.centonze@istitutotumori.mi.it)
Stefano Chiaravalli (stefano.chiaravalli@istitutotumori.mi.it)
Cinzia Paolino (cinzia.paolino@istitutotumori.mi.it)
Massimo Milione (massimo.milione@istitutotumori.mi.it)
Maura Massimino (maura.massimino@istitutotumori.mi.it)
Michela Casanova (michela.casanova@istitutotumori.mi.it)
Patrizia Gasparini (patrizia.gasparini@istitutotumori.mi.it)

Version: 4 Date: 16 May 2018

Author’s response to reviews:

minor revision:

1 - There appears to be one author represented by two sets of initials in the Authors' contributions: SLR and RSL are both present. Please correct one set of initials.

We modified the wrong initials to the right ones SLR.
2 - Please note that the Availability of data and materials section refers to the raw data used in your study and presenting tables and figures is not sufficient to state that all data are contained within the manuscript and additional files. Please only use this statement if you have indeed provided all raw data on which your study is based. We strongly encourage all authors to share their raw data, either by providing it in a supplementary file or depositing it in a public repository and providing the details on how to access it in this section.

Thank you for the observation. Indeed we replaced the statement to a more adequate one for our data in the research article "The datasets used and/or analysed during the current study are available from the corresponding author on reasonable request."

we hope that present form of the manuscript now meets BMC cancer's requirements for publication.

Best regards

Patrizia Gasparini